Author's response to reviews

Title: Are parents' knowledge and practice regarding immunization related to pediatrics’ immunization compliance? A mixed method study

Authors:

Omer Qutaiba B. Allela Dr (omarallela2010@gmail.com)
Mohd B. Bahari (wakbaidi@yahoo.com)
Harith K Alqazaz (harithqazaz73@yahoo.com)
Muhammad R.M. Salih (muhanad_rmk@yahoo.com)
Ramadan M Elkalmi (edriph@gmail.com)
Shazia Q Jamshed (Pharmacist1992@live.com)

Version: 5 Date: 14 January 2014

Author's response to reviews: see over
Authors’ Responses to Editor

We are very thankful for such constructive comments which have significantly enhanced our manuscript entitle “Are parents' knowledge and practice regarding immunization related to pediatrics’ immunization compliance? A mixed method study”

Editor comments:
We are pleased to inform you that the editors have now decided to accept your manuscript. However, before we can proceed, we will need you to make some changes to your manuscript. We strongly urge you to make these changes promptly, as we cannot proceed to the next process until we have received a version containing the changes.

Author response: We are very thankful for you and agree to respond to the editors’ comments.

Author’s action: Changes made as suggested.

Sincerely,

The Corresponding Author

Dr. Omer Q.B. Al-Lela
Assistant Professor in Clinical Pharmacy
School of Pharmacy
Faculty of Medical sciences
University of Duhok (UOD)
Duhok / Iraq
<table>
<thead>
<tr>
<th>#</th>
<th>Comments of editor</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Editor</td>
<td>We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service.</td>
<td>This research is one part from my PhD thesis and already this research sent to professional editing service in United Kingdom (fluent English). And this is the link for editing service We sent it to Proof reading in UK <a href="http://www.proof-reading-service.com/how-it-works.html">http://www.proof-reading-service.com/how-it-works.html</a></td>
</tr>
<tr>
<td>Editor</td>
<td>We would ask that you please state clearly that the protocol was approved by an ethics committee of the Ministry of Health, as this is not currently clear</td>
<td>We do agree with the editor’s suggestion; and changes were made accordingly as suggested</td>
</tr>
</tbody>
</table>